Author’s response to reviews

Title: Clinical Application of Large Channel Endoscopic Decompression in Posterior Cervical Spine Disorders

Authors:

chengli li (315828686@qq.com)
Xiaojie Tang (huyuningX@163.com)
song chen (yantaixulin@126.com)
Yongchun Meng (Yongchun.meng@hotmail.com)
Wei Zhang (byytcl@sina.com)

Version: 5 Date: 10 Oct 2019

Author’s response to reviews:

Thank you for your letter.

Regarding the questions raised by the reviewers, the replies are as follows:

1. The full text for the abbreviation 'CSA' for the first appearance in the abstract has been provided.
2-4. Corresponding errors have been corrected as required.
5. The table has been reset,
   (1). The full text for the abbreviation in the tables has been provided.
   (2). The ' mean number ± standard deviation (±s, n=32).' has been put in the footnotes.
   (3). Corresponding errors have been corrected as required, including other similar mistakes in the article.
6-7. Corresponding errors have been corrected as required.